Author’s response to reviews

Title: Power of computed-tomography-defined sarcopenia for prediction of morbidity after pancreaticoduodenectomy

Authors:

Nicolas Linder (nicolas.linder@medizin.uni-leipzig.de)
Alexander Schaudinn (alexander.schaudinn@medizin.uni-leipzig.de)
Katharina Langenhan (katharina.langenhan@medizin.uni-leipzig.de)
Felix Krenzien (felix.krenzien@charite.de)
Hans-Michael Hau (HansMichael.Hau@medizin.uni-leipzig.de)
Christian Benzing (benzing@charite.de)
Georgi Atanasov (atanasov@charite.de)
Moritz Schmelzle (schmelzle@charite.de)
Thomas Kahn (kahn@medizin.uni-leipzig.de)
Harald Busse (harald.busse@medizin.uni-leipzig.de)
Michael Bartels (michael.bartels@helios-gesundheit.de)
Ulf Neumann (uneumann@ukaachen.de)
Georg Wiltberger (gwiltberger@ukaachen.de)

Version: 2 Date: 08 Apr 2019

Author’s response to reviews:

Dear BMC MI, thanks for the minor revision.

Editorial comments were followed and changes made. The last comments by reviewer 1 are a bit challenging, because these changes were made earlier (table with baseline characteristics, numbers as median values with corresponding secription at the bottom of the table). Therefore this maybe was a misunderstanding. Please let me now if any further changes are still required.

Kind regards